

# Citation for published version

Urbano, D., Aparicio, S., Guerrero, M., Noguera, M. & Torrent-Sellens, J. (2017). Institutional determinants of student employer entrepreneurs at Catalan universities. Technological Forecasting and Social Change, 123(), 271-282.

# DOI

https://doi.org/10.1016/j.techfore.2016.06.021

# **Document Version**

This is the Accepted Manuscript version. The version in the Universitat Oberta de Catalunya institutional repository, O2 may differ from the final published version.

# **Copyright and Reuse**

This manuscript version is made available under the terms of the Creative Commons Attribution Non Commercial No Derivatives licence (CC-BY-NC-ND) <u>http://creativecommons.org/licenses/by-nc-nd/4.0,</u> which permits others to download it and share it with others as long as they credit you, but they can't change it in any way or use them commercially.

# Enquiries

If you believe this document infringes copyright, please contact the Research Team at: repositori@uoc.edu





# **Durham Research Online**

#### Deposited in DRO:

08 December 2017

#### Version of attached file:

Accepted Version

#### Peer-review status of attached file:

Peer-reviewed

#### Citation for published item:

Urbano, D. and Aparicio, S. and Guerrero, M. and Noguera, M. and Torrent-Sellens, J. (2017) 'Institutional determinants of student employer entrepreneurs at Catalan universities.', Technological forecasting and social change., 123 . pp. 271-282.

#### Further information on publisher's website:

https://doi.org/10.1016/j.techfore.2016.06.021

#### Publisher's copyright statement:

C 2016 This manuscript version is made available under the CC-BY-NC-ND 4.0 license http://creativecommons.org/licenses/by-nc-nd/4.0/

#### Additional information:

#### Use policy

The full-text may be used and/or reproduced, and given to third parties in any format or medium, without prior permission or charge, for personal research or study, educational, or not-for-profit purposes provided that:

- a full bibliographic reference is made to the original source
- a link is made to the metadata record in DRO
- the full-text is not changed in any way

The full-text must not be sold in any format or medium without the formal permission of the copyright holders.

Please consult the full DRO policy for further details.

# Institutional determinants of student employer entrepreneurs at Catalan universities

**David Urbano** (david.urbano@uab.cat) (Corresponding author) Universitat Autònoma de Barcelona, Department of Business, Edifici B, Campus UAB, 08913 Bellaterra (Barcelona), Spain.

## **Sebastian Aparicio**

Universitat Autònoma de Barcelona, Department of Business, Edifici B, Campus UAB, 08913 Bellaterra (Barcelona), Spain; and Fundación ECSIM, Medellín-Colombia.

## **Maribel Guerrero**

Deusto Business School, University of Deusto and Orkestra-Basque Institute of Competitiveness Mundaiz 50, 20012 Donostia-San Sebastian, Spain

## Maria Noguera

Universitat Autònoma de Barcelona, Department of Business, Edifici B, Campus UAB, 08913 Bellaterra (Barcelona), Spain.

## Joan Torrent-Sellens

Universitat Oberta de Catalunya, Economics and Business Studies, and Internet Interdisciplinary Institute (IN3) MediaTIC Building, Roc Boronat Street, 117, 6th Floor, 08018 Barcelona, Spain.

# Abstract

This article investigates the influence of informal and formal institutions on the university students' decision of becoming employer entrepreneurs in the context of Catalonia. A sample of 1,207 students from the Universitat Autònoma de Barcelona (UAB) and the Universitat Oberta de Catalunya were surveyed for the period of 2012-2015, and probit regressions over pooled data were used. The main findings suggest that formal factors (university's lack of incentives to create a new business, entrepreneurial knowledge, training and skills, and entrepreneurship education) are higher correlated with the student employer entrepreneurs than informal institutions (role models, entrepreneur's social image and fear of failure). Despite that the entrepreneur's social image does not seem to have an influence on the entrepreneurial decision of university students, the other variables analysed are statistically significant, correlated with entrepreneurship as a choice. Specifically, entrepreneurship education is the most relevant variable in explaining the decision of university students becoming employer entrepreneurs. The paper contributes with policy discussions in order to extend the current debate about the role of the universities in the entrepreneurial process and also the importance of entrepreneurial universities to the society.

**Keywords:** Student employer entrepreneurs, entrepreneurial universities, institutions, Catalonia.

# 1. Introduction

Since Schumpeter (1911), there has been an explicit and generalized recognition of the role played by the entrepreneurs in generating economic development. According to Audretsch (2012), in the last decades a growing explosion of articles providing theoretical and empirical evidence about the effect of entrepreneurship has occurred. In this respect, entrepreneurial activity has been linked to economic growth (Audretsch and Keilback, 2008; Wennekers and Thurik, 1999; Urbano and Aparicio, 2015), as well as cluster formation (Lee et al., 2012; Li et al., 2015; Rocha, 2004) and new jobs creation (van Praag and Verslot, 2007; van Stel et al., 2014b), among other effects. According to van Praag and Verslot (2007), entrepreneurship contributes to society by including individuals in the labour market, sharing with them a common project.

Although literature dealing with the entrepreneurial decision of individuals in creating jobs exists (Millán et al., 2012; Millán et al., 2015; Thurik et al., 2008; van Stel et al., 2014a,b, among others), most of the works are focused on understanding the individual characteristics conditioning the probability of become an entrepreneur. However, recent evidence has suggested that other non-individual factors could affect this probability. For instance, van Stel et al. (2014a) have found that the technological environment affects the probability of becoming an employer entrepreneur. These authors define this category as those entrepreneurs creating new firms that employ personnel, contributing therefore to the growth and development process. In line with the technological environment analysis, science and technology are considered some of the main sources of economic and social development at both regional and national levels (Aghion and Howitt, 1992). Presumably, the development is obtained through technical and social innovations brought to the market via knowledge mechanisms (Karaka and Öner, 2015). In this respect, universities play an important role by providing this innovative environment, which is widely recognized. For example, Guerrero et al. (2015) found that entrepreneurial university, through the knowledge spillover and creation, could impact the economic growth. Also, Etzkowitz and Leydersdorff (1995, 2000) and Etzkowitz et al. (2000) suggest that the enhanced role of universities could increase the knowledge-based societies. Thus, given the universities' function creating and consuming new knowledge, their role in social value creation has gained attention in the sense that universities could be promising scenarios that encourage the worth entrepreneurial activity, and therefore, they could be viewed as an important policy mechanism influenced by the institutional factors (Guerrero and Urbano, 2012). Although there are studies analyzing those institutions affecting the development of entrepreneurial universities and their subsequent influence on socioeconomic performance (Kirby, 2006; Kirby et al., 2011; Guerrero et al., 2014, 2015; Guerrero and Urbano, 2012), few studies have considered how institutional environment influences students to become entrepreneurs by hiring

employees and including them into a common project (Urbano and Guerrero, 2013). Here, further studies could extend the current debate about the role of the universities in the entrepreneurial process and also the importance of entrepreneurial universities to the society.

Therefore, this article investigates the influence of informal and formal institutions on the university students' decision of becoming employer entrepreneurs in the context of Catalonia. By using institutional economics (North, 1990, 2005), we consider informal (role models, entrepreneur's social image and fear of failure) and formal institutions (the university's lack of incentives to create a new firm, entrepreneurial knowledge, training and skills and entrepreneurship education). A sample of 1,207 students from the Universitat Autònoma de Barcelona (UAB) - onsite university - and the Universitat Oberta de Catalunya – online university – were surveyed for the period of 2012-2015 and probit regressions over pooled data were used. Throughout probit models, we predict the probability of student entrepreneurs hiring between 1 and 50 workers or at least one employee and also more than 50 employees. Our findings show that the formal factors (university's lack of incentives to create a new firm, entrepreneurial knowledge, training and skills, and entrepreneurship education) have greater effect on the probability of becoming a student employer entrepreneur than informal institutions (role models and fear of failures). Among the formal factors, entrepreneurship education is found to have the greater impact in explaining the entrepreneurial decision of university students.

The remainder of the paper is as follows. Section 2 examines the literature on institutions and entrepreneurship in terms of university students. Section 3 then describes the data collection and the methodology used to empirically examine the influence of institutions on student employer entrepreneurs. Section 4 presents the results and the robustness check, and section 5 discusses some of the policy implications related to the significance of entrepreneurial universities and their contribution to society. In the final section, the paper concludes and suggests future research lines.

# 2. Theoretical framework: Institutional economics and entrepreneurial universities

As noted before, we base our analysis upon the institutional economics framework (North, 1990, 2005). In this regard, North (1990) defines the institutions as those rules shaping the human interactions and economical decisions that create specific market qualities. These rules can be either informal institutions, such as the culture, values and the social norms of a particular society, or formal, such as regulations, contracts, procedures and so on. Informal rules arguably tend to reduce the uncertainty originated by the complex and dynamic interactions between the individuals; meanwhile, formal institutions are aimed to reduce transaction costs based on regulations (North, 1990, 2005). Here, depending on whether those interactions are complex and how the individual intentionality toward progress is, higher regulative efficiency could be achieved or not, which improves or deters the

social norms that create progress intentionality. Thus, informal institutions constrain the nature of formal institutions and vice versa. In terms of time, formal institutions can change in the short-term, whereas informal institutions tend to change in the long-term (Williamson, 2000).

Building upon the institutional economics, some scholars have been interested in understanding how the entrepreneurial activity is framed by institutions (Aidis et al., 2008; Bruno et al., 2013; Bruton et al., 2009, 2010; Jennings et al., 2013; Thornton et al., 2011; Urbano and Alvarez, 2014; among others). In this respect, both informal institutions (cultural and social norms, perception of start-up opportunities and the entrepreneur's social image, among others) and formal rules (intellectual property rights, procedures, laws and so on) have been found to be influential factors in entrepreneurship (Aparicio et al., 2015; Autio and Fu, 2015; Knörr et al., 2013; Maas et al., 2014; Stenholm et al., 2013; Stephan et al., 2015; Turró et al., 2014; van Hemmen et al., 2015; Zhang, 2015; among others). Recently, an increasing amount of works have recognized and provided additional empirical evidence about the significant effect that institutional factors, in addition to other variables, specifically have on student entrepreneurs and on the creation and development of entrepreneurial universities in general (Guerrero et al., 2014; Guerrero et al., 2015; Guerrero and Urbano, 2012; Urbano and Guerrero, 2013). In this sense, Guerrero and Urbano (2012) highlight that student entrepreneurs socialize according to informal institutions (e.g., role models, university community's attitude toward entrepreneurship and reward system), as well as formal factors, such as university governance structure, support measures for entrepreneurship and entrepreneurship education. Here it has been argued that knowledge-based entrepreneurship, emerging from both the institutional norms outside the university and those university's laws, structures and supports toward entrepreneurship, is quite relevant to achieve higher levels of job creations, competitiveness and economic growth (Guerrero et al., 2015; Urbano and Guerrero, 2013).

In terms of informal institutions, among other authors, Aidis et al. (2008), Estrin and Mickiewicz (2012), Noguera et al. (203), Stephan et al. (2005) and van Hemmen et al. (2013), have suggested that role models, entrepreneur's social image and fear of failure, among other factors, affect the entrepreneurial activity, urging or discouraging its development process. In general, there is a vast amount of literature arguing the importance of these variables, as those institutions highly correlated with the entrepreneurial activity. For instance, Álvarez et al. (2014) present a literature review identifying that role models, entrepreneur's social image and fear of failure are vastly explored within the entrepreneurship research. Likewise, Urbano and Alvarez (2014) outline these three elements, among others, as the most relevant due to their proximity to the informal institutions. Taking into consideration the importance of these institutions, Guerrero and Urbano (2012) proposed an integrated model to understand how role models, rewards system and specific attitudes toward entrepreneurship explain the inner dynamics of entrepreneurial universities. According to Guerrero et al. (2014), these informal institutions could define the creation of new ventures among the researchers, professors and university students.

In this sense, we are attempting to evaluate whether or not the already-identified institutions affect the student employer entrepreneurs.

Specifically, regarding the role models, it has been argued that people might reduce the level of uncertainty and have more confidence, since they know and learn from other entrepreneurs, which in turn raises the likelihood of starting a new venture (Davidsson and Honig, 2003). For example, some authors suggest that children with entrepreneur parents are likely to have worked with them at an early age, which allows children gaining experience and attitudes, useful later to start their own business (Carroll and Mosakowski, 1987; Hoffmann et al., 2015; Lindquist et al., 2015; Scott and Twomey 1988; Scherer et al. 1989; van Auken et al. 2006). Role models from family and labour experience lead to the perception of oneself as an entrepreneur (Scott and Twomey, 1988; Zapkau et al., 2015). Scherer et al. (1989) argue that entrepreneurial role models are a property of a high number of entrepreneurs. Van Auken et al. (2006) and Hoffmand et al. (2015) show that many business owners include their family members like sons and daughters, among others, in their businesses, generating interaction and involvement of individuals within the new firm, which means the greatest impact on entrepreneurial intentions in those family members. Moreover, some of the literature focused on the intentions suggests that the family background affects positively the entrepreneurial attitudes (Kolvereid, 1996; Krueger, 1993; Maes et al., 2014; Matthews and Moser, 1995; Scherer et al., 1989; among others). Thus, the presence of entrepreneurs with labour experience in family business and successful role models transmit positive effects to potential entrepreneurs (Gnyawali and Fogel, 1994). Likewise, Arenius and Minniti (2005) posit that different social sciences recognize the importance of knowing other entrepreneurs for business creation choice. For example, in psychology, some authors have discussed the importance of role models because of their ability to enhance selfefficacy (Baron, 2000; Begley and Boyd, 1987). Minniti (2005), in economics, has discussed increases in individuals' confidence generated by the presence of knowing other entrepreneurs and the individuals' ability to reduce ambiguity. In sociology, for example, Aldrich (1999) has discussed the role of personal networks and their ability to enhance entrepreneurial confidence by providing advice, support and examples. In terms of university scientists and students, Guerrero and Urbano (2012), Guerrero et al. (2014) and Kirby et al. (2011), among others, have discussed and provided evidence about the importance of role models for these kinds of entrepreneurs. Likewise, Venkataraman (2004) has suggested that forums and meetings often occur in the university environment, and these activities open opportunities for entrepreneurial role models to exert informal influence that might positively influence the entrepreneurial activity in students. Finally, we propose the following hypothesis:

**H1.** Role models have a positive effect on the probability of university students becoming employer entrepreneurs.

Concerning the entrepreneur's social image, authors such as Weenekers and Thurik (1999) and Acs et al. (2013) address the debate toward the importance of entrepreneurs not only for economic but also for social value creation. Recently,

Laužikas and Dailydaitė (2015) have related the entrepreneur's social image with the social capital created by the entrepreneurs. If entrepreneurs, through their actions (i.e., innovations, productivity and employment generation, etc.), reflect their success in the society, they gain trust and good reputation, and therefore, social networks increase (Carter et al., 1998). Alvarez et al. (2011b) have explored the entrepreneurial activity in Spanish regions, where the informal institution, such as the entrepreneurial social image, is positive and statistically significant correlated with the entrepreneurial activity. Similarly, Jayawarna et al. (2013), throughout cluster analysis, identified that prestige is a useful element required by the business life course, especially for early establishments. Parker and van Praag (2010) and van Praag (2011) posited that depending on the status generated by the entrepreneurs, other individuals would tend to choose the entrepreneurship as a professional career. Particularly, van Praag (2011) has found that the entrepreneur's status is positively associated with the probability of an individual becoming an entrepreneur. In this respect, the way in which the entrepreneur is perceived within the society encourages this kind of initiatives in the individuals. Additionally, it is also found that the entrepreneur's prestige not only encourages the entrepreneurial activity but also its survival across time (Chou et al., 2013). In this regard, Warnecke (2014) has argued that the design of some public policies dealing with entrepreneurship explained their efficacy. Warnecke's work is useful to understand how the trust gained (or lost) by the entrepreneurs has led to creation of microfinance strategies, which are specially needed by those young entrepreneurs. However, the trust-building process seems not to be enough for those student entrepreneurs. Although Loras and Vizcaino (2013), for a sample of engineering students, have found positive evidence concerning the effect of entrepreneur's social image on the decision to become an entrepreneur, their results were not statistically significant, suggesting that despite the entrepreneurial recognition, potential engineering entrepreneurs prefer to look to work for established firms. Still, from the theoretical perspective, the entrepreneur's social image is expected to positively influence the entrepreneurial activity. Here, Nielsen and Lassen (2012) have provided a thorough literature review on the image of entrepreneurs who are facing problems such as opportunity recognition, complex evolution, contingencies and social challenges. Hence, we posit the following hypothesis:

**H2.** A favourable entrepreneur's social image has a positive effect on the probability of university students becoming employer entrepreneurs.

In terms of fear of failure, it is argued that individuals are averse to risk, and therefore, the perceived possibility of failure is an important component of an individual's decision to become an entrepreneur, affecting the choice to set up a new firm (Minniti and Nardone, 2007). Langowitz and Minniti (2007) suggest that risk exists when individuals make decisions about financial support and market explorations, among other decisions. Here, to some degree, fear of failure affects the behaviour of individuals. Business creation *per se* has risk in the overall process (i.e., ideas exploration, evaluation and exploitation). In this sense, Weber and Milliman (1997) claim that if individuals are the most risk averse, it is likely to reduce the level of

starting a new business. Therefore, fear of failure can reduce the alertness to perceive some business ideas, and hence, it limits the evaluation capacity of opportunity exploitation. Nevertheless, Cacciotti and Hayton (2015) suggest that there could be dualistic nature of this factor since previous failure experiences could be positive for entrepreneurs facing difficulties and high-risk decisions. The recent and thorough work of Cacciotti and Hayton concludes that the fear of failure is defined as contextspecific phenomenon that most of the literature reviewed considers as a barrier for entrepreneurs. A cross-country analysis at individual level. Urbano and Alvarez (2014) provide empirical evidence about the effect of fear of failure on the entrepreneurial activity. These latest findings are highly relevant, due the country specification. Here, Urbano and Alvarez (2014) suggest that this factor is common worldwide. Similarly, Wennberg et al. (2013) find that the relationship between fear of failure and entry is moderated by cultural practices of institutional collectivism and uncertainty avoidance. At regional levels, Mancilla and Amorós (2015) provide evidence regarding the differences of the fear of failure between women and men. Accordingly, female entrepreneurship tends to suffer greater negatives than their male counterparts (Koellinger et al., 2013: Noguera et al., 2013). In terms of entrepreneurial students, Lukes and Souhar (2015) have found that even though fear of failure deters the probability of becoming entrepreneurs, there is a lower effect of this variable compared to non-students, since the university context easily allows the perception of new opportunities. Still, the above literature vastly suggests the negative impact of fear of failure on the probability to become an entrepreneur, especially if those new entrepreneurs face not only risk for them but also for their workers (van Stel et al., 2014a). Thus, the hypothesis proposed is:

**H3.** Fear of failure has a negative effect on the probability of university students becoming employer entrepreneurs.

Concerning formal institutions, it is also well documented that entrepreneurs socialize in particular contexts, which are under regulative norms and government structures (Aparicio et al., 2015; Djankov et al., 2002; Guerrero and Urbano; 2012; Kirby et al., 2011). With our particular concern, Wach (2014) has studied the evolution of the entrepreneurship education policy in Europe. According to this author, although since the 1980s, the importance of entrepreneurship for social development has been recognized, in the early 2000s a specific policy was designed to promote entrepreneurship in higher education, where universities must include a specific subject dealing with entrepreneurial knowledge, training and skills. In this regard, an important part of the curriculum across the undergraduate (and also graduate) programmes should be focussed on entrepreneurship. Here it is expected a growing intention toward entrepreneurship based on opportunities recognition. Additionally, political structure of universities should be in accordance with these policies. Wach (2014) finds that an incentive mechanism is also proposed in the European entrepreneurship policies. Accordingly, universities should be enabling professors, researchers and students to achieve outcomes related to the entrepreneurial activity. thus allowing the knowledge transfer to the society (Guerrero et al., 2014).

In the above respect, the formal institutions of the university context are argued to frame student entrepreneurs (Guerrero and Urbano, 2012). In this regard, special emphasis has been given regarding the role of universities in development construction nowadays (Trippl et al., 2015). Universities should not only serve as the actors' articulator (e.g., innovation systems and incumbent firms), but also as an environment encouraging the entrepreneurial activity among the students. In this sense, university incentives toward the entrepreneurial activity could define the intentionality of their potential entrepreneurial students. Following Guerrero et al. (2008), Kirby (2006) and Wright et al. (2007), an accurate system of incentives represents strategic actions aimed to promote entrepreneurship in two ways: monetary (scholarships, fellowships, etc.) and non-monetary (contests, awards, etc.). Additionally, the work of Rendisch et al. (2015) supports that the universities incentives toward the alliances between the university and incumbent firms is a constructive exchange of experience and knowledge useful to the regional growth and the competitiveness. Here these strategies could be the key point to attract student employer entrepreneurs and to recruit highly qualified employees. However, according to Audretsch et al. (2014) and Audretsch et al. (2012), it is possible that the competitiveness gained from the knowledge spillover causes a crowding-out effect for those new entrepreneurs. In this respect, the work of Audrestsch et al. (2014) suggests that the university incentives should take into account these kinds of side effects, since the established firms of professors and researchers do not allow the development of the new firms in younger entrepreneurs. Here it is also possible that the lack of incentives or inappropriate strategies prove harmful for the entrepreneurial intentions of university students. Therefore, the following hypothesis is proposed:

**H4.** The universities' lack of incentives has a negative effect on the probability of university students becoming employer entrepreneurs.

Regarding the entrepreneurial knowledge, training and skills, Estrin and Mickiewicz (2012) have argued that these elements are highly important in the decision to set-up a new businesses, in particular those requiring a high level of knowledge. According to Harper (2003), entry decisions are conditional on individual skills as well as on the national economic context. Some skills related to new business creations are powered by the self-confidence of each entrepreneur (van Hemmen et al., 2013) and the university environment where they belong. According to the work of Hemmen et al., education and skills could promote positive interaction between the groups that form each new firm based on opportunity. With respect to the benefits for each entrepreneur, Harper (2003) suggests that particular training and skills in entrepreneurship enhance the feelings of internal control and personal agency, which at the same time promote the alertness in entrepreneurs. This alertness sensitivity leads to opportunity perception with a lower level of uncertainty. Thus, entrepreneurial education positively affects the capacity to create new businesses by opportunity with a higher potential for growth. In this sense, Estrin and Mickiewicz (2012) find empirical evidence about the impact of some skills on entrepreneurship. Additionally, Guerrero et al. (2014) have found that students enrolled in a university

with entrepreneurial characteristics increase their possibility to detect opportunities in the labour market. In essence, particular characteristics of entrepreneurship programmes within the universities, such as the orientation to increase the university students' attitudes/values, enhanced managerial skills and developed networks that are detected to be beneficial to those future employer entrepreneurs. To transform the entrepreneurial mindset of student entrepreneurs, a balance between knowledge, skills and training is necessary. In line with Moog et al. (2015), it is likely that particular skills and training positively affect the decisions of university students to become entrepreneurs. Kolstad and Wiig (2015) find that a variety of skills not only affect the entrepreneurial intention but also the entrepreneurial success, which means that entrepreneurs are growing and constantly creating jobs. Thereby, we propose the following hypothesis:

**H5.** Entrepreneurial knowledge, training and skills have a positive effect on the probability of university students becoming employer entrepreneurs.

Regarding an entrepreneurship education policy from the university, the literature provides examples of its importance for those potential student entrepreneurs. For instance, Link and Scott (2005) and Grandi and Grimaldi (2005) suggest instruments such as the support of centres of small-university businesses, research facilities, research groups, technology transfer strategies and incubators, in order to provide entrepreneurial experience (e.g., through learning-by-doing process) to enrolled students. This could increase the knowledge of how small firms operate, the identification of opportunities and the development of networks, among other increases (Link and Scott 2005; Grandi and Grimaldi 2005). In this regard, Fayolle (2007a,b) suggests that an adequate educational programme could provide a wide variety of situations, aims, and methods oriented toward improving entrepreneurial skills and attributes of enrolled students who could develop both creative and critical thinking. Also, Guerrero and Urbano (2012), Louis et al. (1989) and Liñán et al. (2011), among others, argue that entrepreneurship education from the beginning of undergraduate programmes could enhance the students' attitudes toward the entrepreneurial activity and could favourably facilitate the development of potential entrepreneurs at all university levels. These important university policies affecting students' entrepreneurial activity allow the recognition of those potential entrepreneur agents who identify opportunities, mobilize resources, and constantly create social and economic value (Degroof and Roberts, 2004). Hence, the university policies about entrepreneurship education should be in line with those institutions building capacities toward sustainable regional and national economic development (Feldman, 2014). By doing this, entrepreneurs obtain benefits from these locations in terms of support, resources, contacts and so on. In terms of the university courses, the main benefits obtained from topics such as business plan, business model, marketing strategy, among others, not only increase entrepreneurial intention but also improve the learning of their future jobs (Chen et al., 2015). Additionally, the entrepreneurship education at tertiary level increases the official formation of new businesses and decreases the unofficial activity (Jiménez et al., 2015). Overall, given that the university student entrepreneurs are also pivotal agents of change that can transform

the regional and national development, the universities and their entrepreneurship education system contribute indirectly to the positive dynamics of jobs creation, competitiveness and socioeconomic development (Guerrero et al., 2015). Taking into account the previous discussion, the following hypotheses is posited:

**H6.** Entrepreneurship education has a positive effect on the probability of university students becoming employer entrepreneurs.

# 3. Methodology

## 3.1. Data and variables

Two Catalan universities were selected to analyse the influence of informal and formal institutions on the university students' decisions of becoming employer entrepreneurs. It is worth mentioning that Catalonia is considered to be a main Spanish industrial centre. According to the Spanish Statistical Office (Instituto Nacional de Estadística, INE), Catalonia produces 18.6% of the total Spanish GDP and creates 17.8% of total employment, with high productivity dynamics. In 2014, it represented 16.09% of the total Spanish population, with an income level above the national average (118%). In terms of its entrepreneurial activity, the Catalan number of existing and newly created firms is approximately 20% of the total number of firms in Spain.

Catalonia counts as a university system with nine universities well positioned in the European context within the high top rankings (i.e., QS World University Rankings, Shanghai Academic Ranking of World Universities, etc.). Additionally, the geographical localization, as well as the high quality educational level, has made Catalonia an attractive place to students worldwide. According to Espinet et al. (2015), despite the civil war suffered in Spain during the middle of the twentieth century, the Catalan educational system was consolidated in the 1970s and 1980s. Through this consolidation process, some Catalan universities have emerged in order to extend the coverage of young and adult populations. Hence, the Universitat Autònoma de Barcelona (UAB) and the Universitat Oberta de Catalonia (UOC) were created in 1968 and 1995, respectively, with the purpose to offer a wide range of disciplines (i.e., humanities, social sciences, health sciences, experimental sciences, etc.). However, both of them are differently characterized. For instance, on the one hand, UAB has a very large campus, where most the academic and research activities are developed, making it a favourable environment to innovate and undertake. In this regard, UAB has 226 research groups constantly producing patents, research papers and spin-offs. On the other hand, though UOC is an online university, it also has an important research production. For instance, UOC has 46 research groups contributing with articles, books, research projects and spin-offs. Here, it is worth mentioning that UOC contains one technology transfer office that supports its innovative and entrepreneur students. The main characteristics of both universities are summarized in Table 1.

Description		UAB <sup>a</sup>	UOC <sup>b</sup>				
General information	Focus	Humanities, Social Sciences, Health Sciences, Experimental Sciences and Engineering.					
	Year of foundation	1968	1995				
	Nature	Public University	Public University				
	University size	28012 undergraduates	52513 undergraduates				
	University mode	Onsite	Online				
	No. of Professors and researchers	3571 (Staff)	324 (Staff); 3022 (Collaborators)				
Technical details	Criteria	University students enrolled in Economics and Business Administration					
	Population	2634 <sup>c</sup>	4962 <sup>d</sup>				
	Sample size	629	578				
	Sample error	2.	84%				
	Confidence interval	95% ( <i>Z</i> = 1.	96, $p = q = 0.5$ )				
<sup>a</sup> Source: http://www.u	ab.cat/web/conoce-la-uab-cei/la-ua	ab-en-cifras-1345668682739.html (Accessed	l 6 September/2015).				

#### Table 1. Main details of surveyed universities

<sup>b</sup> Source: http://www.uoc.edu/portal/es/universitat/coneix/fets-xifres/index.html (Accessed 6 September/2015).

<sup>c</sup> Source: https://www.uab.cat/web/la-facultad/la-facultad-en-cifras/estudiantes-1345672798019.html (Accessed 6 September/2015).

<sup>d</sup> Source: http://www.uoc.edu/portal/es/uoc-news/actualitat/2014/noticia\_163/graduacio-barcelona-2013-2014.html (Accessed 6 September/2015).

Given the previously mentioned characteristics, in this paper we analyse the student entrepreneurs of these two universities. Thus, empirical data for this research was obtained from a total population of 7,596 university students, during the period of 2012–2015 (2,634 students from UAB and 4,962 students from UOC; the last year is covered until June). The empirical analysis has been carried out upon a sample of 1,207 university students from the UAB (629) and UOC (578). This is a convenience sample very often used in entrepreneurship research (Fayolle and Gailly, 2005; Kolvereid, 1996; Krueger et al., 2000; Liñán et al., 2011; Tkachev and Kolvereid, 1999; Veciana et al., 2005).

In particular, the data were obtained from the Observatory for University Entrepreneurial Activity (Observatori d'Emprenedoria Universitària – OBSEU) (Alvarez et al., 2011a; Urbano et al., 2015), whose main instrument to gather the information was the previously tested questionnaire on the attitudes toward the new firm's creation. Questionnaires were promoted optionally online from the Centre for University Entrepreneurship (CIEU, Centre d'Iniciatives Emprenedores Universitàries - UAB), which is part of UAB Emprèn and collaborates with the Entrepreneurial University Network (Xarxa d'Emprenedoria Universitària -XEU). Those students who were enrolled for business and economics degrees, with previous professors' authorization, were questioned. In that context, response rate was very high. Only a small number of them were incomplete or lacked consistency, and were therefore rejected. Few others had a small proportion of missing data, but they were always less than 2% of items. They were therefore retained. Given the final usable questionnaires. the sample error obtained was  $\pm 2.84\%$  at a 95% confidence level (Z = 1.96, p = q = 0.5). Nevertheless, questionnaires with missing data have been left out for the specific analysis of some items in which data were lacking. Additionally, the instrument was carefully crosschecked with those instruments used by other researchers, such as Chen et al. (1998), Kickul and Zaper (2000), Kolvereid (1996), Kolvereid and Isaksen (2006), Krueger et al. (2000) or Veciana et al. (2005). Ajzen's (1991, 2001, 2002) work was useful to carefully revise and solve any discrepancy that could arise during the questionnaire-making process, compared with other instruments.

Concerning the variables, first, our dependent variables include those university students that have created a new firm and have hired between 1 and 50 employees, and those who have created at least one job and also more than 50 jobs. Following van Stel et al. (2014a), we pretend to measure the probability of those student entrepreneurs that have the intention to create employment, taking into account or not the new firm size. Concerning the informal factors, binary variables that capture if students know entrepreneurs in their families (role model), if there is a positive acceptance of the entrepreneurial activity within their close environment (entrepreneur's social image), and if their perception of fear of failure is considered as an entrepreneurial barrier were used. In terms of formal institutions, two binary variables regarding if university students have the knowledge, training and skills to carry out an entrepreneurial project, as well as if they consider that they have learned entrepreneurial topics, such as business plan and business model within the university's

lack of incentives to create new business is an obstacle (1 not important; 5 very important) was used.

In terms of control variables, we include some observable heterogeneity regarding the income level, gender and age. According to Carree et al. (2002, 2007), the entrepreneurial activity is related to the country development stage. By analyzing at individual level, it is relevant to relate the income level to the entrepreneurial decision, since it is expected that higher income level explains higher entrepreneurial activity rate. Concerning the gender analysis, Arenius and Minniti (2005) have found that the men tend to have higher entrepreneurship rate than the women. And Urbano and Alvarez (2014) suggest that individuals at certain ages upwards tend to present higher attitudes toward the entrepreneurial activity. Table 2 presents the variables used to carry out our analysis.

Variables	Description			
Dependent variables				
Student employer entrepreneurs (1-50 workers)	Dummy variable that takes the value 1 if university students have created a new firm with employees between 1 and 50; 0 otherwise. Dummy variable that takes the value 1 if university			
Student employer entrepreneurs (1 or more workers)	students have created a new firm with more than 1 employee.			
Independent variables				
Informal institutions				
Role model Entrepreneur's social image	Dummy variable that takes the value 1 if respondents have entrepreneurs in their families; 0 otherwise. Dummy variable that takes the value 1 if respondents have a favourable perception about success entrepreneurs in their closer environment; 0 otherwise.			
Fear of failure	Dummy variable that takes the value 1 if respondents manifest fear of failure as obstacle to undertake.			
Formal institutions				
University's lack of incentives to create a new firm	Lack of incentives to create a new firm at university for students entrepreneurs. Scaled variable ranged from 1 (is not important) and 5 (is very important).			
Entrepreneurial knowledge, training and skills	Dummy variable that takes the value 1 if respondents have aquired knowledge and entrepreneurial habilities from university; 0 otherwise.			
Entrepreneurship education	Dummy variable that takes the value 1 if respondents have developed business plan preparation at the university; 0 otherwise.			
Control variables				
Income less than 1000€	Dummy variable that takes the value 1 if respondents have delcared less than 1000€ as income level; 0 otherwise.			
Income between 2000 and 4000€	Dummy variable that takes the value 1 if respondents have delcared an income level between 2000 and 4000€; 0 otherwise.			

## Table 2. Variables description

Variables	Description			
	Dummy variable that takes the value 1 if respondents			
	have delcared an income level between 4000 and 7000€;			
Income between 4000 and 7000€	0 otherwise.			
Gender	Dummy variable that takes the value 1 if respondent is man; 0 otherwise.			
Age	Variable created according to the difference between 2015 and the year of birth.			

#### 3.2. Methods

In order to provide a framework for the empirical analysis, standard binary probit models were used. Thus, as usual, the probability of switching from the starting status to the final is assumed to depend on a set of observed informal and formal institutions, as well as on individual characteristics (as control variables), *IIt*, *FIt* and *Xt*, respectively. Thus, a university student who is either a non-entrepreneur or entrepreneur without employees at time *t* will be observed in student entrepreneur employer (with 1 and 50 workers or at least 1 worker and also more than 50 workers) at time *t* if the utility derived from becoming entrepreneur creating employment (student employer entrepreneur, SEE) exceeds that obtained from being a regular student, paid employee or entrepreneur without workers (Other). Consequently, the probability of switching can be written as:

$$\Pr(Y_{i,t} = 1) = \Pr(SEE_{i,t} = 1 | II_{i,t}, FI_{i,t}, X_{i,t} = 0)$$
  
= 
$$\Pr(U_{i,t}^{SEE_{1-50} \text{ or } SEE_{>1}} > U_{i,t}^{Other} | U_{i,t}^{SEE_{1-50} \text{ or } SEE_{>1}} \le U_{i,t}^{Other})$$
  
= 
$$F(\gamma' II_{i,t} + \delta' FI_{i,t} + \beta' X_{i,t} + \varepsilon_{i,t}) (1)$$

where  $Y_{i,t} = 1$  if the university student becomes employer entrepreneur (with 1 and 50 workers, *SEE*<sub>1-50</sub>; or at least 1 worker and also more than 50 workers, *SEE*<sub>>1</sub>) in period t, and  $Y_{i,t} = 0$ , if the individual continues as student, paid employee or entrepreneur without employees in period t. For this exercise,  $\gamma$ ,  $\delta$  and  $\beta$  are the associated vectors of informal institutions ( $II_{i,t}$ ), formal institutions ( $FI_{i,t}$ ) and control variable coefficients to be estimated, respectively;  $\mu_{i,t}$  is a disturbance term that includes the time-invariant unobserved heterogeneity, and  $F(\cdot)$  is specified as the normal cumulative distribution function.

For comparison purpose, our estimations are based upon the linear probability model (through ordinal least square, OLS) (Models 1 and 2), a discrete choice model without university fixed-effects (Models 3 and 4) and with university fixed-effects (Models 5 and 6) through probit estimation. The first set models (1, 3 and 5) estimate the probability of a university student becoming an employer entrepreneur, hiring between 1 and 50 employees; whereas the second set of models (2, 4 and 6) assess the probability of a university student becoming an employer entrepreneur, hiring at least 1 employee and also more than 50 employees. Here, our purpose is to explore the differences between those new firms restricted to one specific size and those new ventures no matter their size.

## 4. Results and discussion

Table 3 provides means, standard deviations and pairwise correlation coefficients for the variables we studied. The table shows that in our sample there is on average 9.3% of university students that are employer entrepreneurs hiring between 1 and 50 employees, and 9.8% are student employer entrepreneurs that have at least one worker and also more than 50 employees.

	Variable	Obs	Mean	Std. Dev.	Min	Max	1	2	3	4
1	Student employer entrepreneurs (1-50 workers)	1207	0.093	0.290	0	1	1			
	- Student employer entrepreneurs (1 or more									
2	workers)	1207	0.098	0.297	0	1	0.972*	1		
3	Role model	1207	0.698	0.459	0	1	0.043	0.053*	1	
4	Entrepreneur's social image	1207	0.781	0.414	0	1	0.017	0.019	0.014	1
5	Fear of failure	1206	0.536	0.499	0	1	-0.080*	-0.085*	-0.087*	0.022
6	University's lack of incentives to create a	1207	2 000	0.052	1	F	0.001*	0.060*	0 020	0.002
	Fintrenreneurial knowledge training and	1207	3.990	0.932	1	5	-0.001	-0.009	-0.039	0.005
7	skills	1206	0.498	0.500	0	1	0.139*	0.130*	0.040	-0.059*
8	Entrepreneurship education	1207	0.600	0.490	0	1	0.121*	0.115*	0.015	0.055*
9	Income less than 1000€	1204	0.116	0.321	0	1	0.037	0.047	-0.072*	-0.014
10	Income between 2000 and 4000€	1204	0.411	0.492	0	1	0.031	0.022	-0.021	-0.002
11	Income between 4000 and 7000€	1204	0.140	0.348	0	1	0.028	0.029	0.062*	0.047
12	Gender	1207	0.439	0.496	0	1	0.103*	0.097*	-0.068*	-0.008
13	Age	1207	30.338	8.926	16	70	0.038	0.037	-0.003	-0.038
	Variable	5	6	7	8	9	10	11	12	13
5	Fear of failure	1								
6	University's lack of incentives to create a									
Ū	new firm	0.038	1							
7	Entrepreneurial knowledge, training and skills	-0.150*	-0.013	1						
8	Entrepreneurship education	-0.103*	-0.016	0.182*	1					
9	Income less than 1000€	0.068*	0.019	-0.024	0.016	1				
10	Income between 2000 and 4000€	-0.016	-0.019	0.056*	0.011	-0.303*	1			
11	Income between 4000 and 7000€	-0.018	-0.055*	0.018	0.042	-0.147*	-0.338*	1		
12	Gender	-0.078*	-0.085*	0.155*	0.089*	-0.008	-0.022	0.071*	1	
13	Age	-0.059*	-0.044	0.089*	0.029	-0.042	0.020	-0.031	0.031	1

## Table 3. Descriptive statistics and correlation matrix

\* p < 0.1.

Additionally, the correlation analysis shows several significant correlations that met our expectations in terms of the informal and formal institutions. In order to test for the problem of multicollinearity, we calculated the VIF value for equation (1), which is 1.11. Thus, multicollinearity is not a problem in the analysis. Furthermore, to address the possibility of heteroskedasticity among observations pertaining to the same university, robust standard errors were estimated.

#### 4.1. Main findings

Table 4 shows that all the models are highly significant ( $p \le 0.000$ ). The *F*-statistic confirms that the joint effect of informal and formal institutions, along with the control variables, explains the decision of university students to become employer entrepreneurs. Additionally, the higher adjustment is obtained when university fixed-effects (in our case, a dummy variable representing UOC) is introduced within the model. Thus, Model 5 explains the 9.9% of the variability of those university students becoming employer entrepreneurs with between 1 and 50 workers; while Model 6 explains 8.8% of the variability of those student entrepreneurs without new firm size restriction. Furthermore, the AIC and BIC criteria show that Models 5 and 6 are higher relevance than Models 1-4.

	Linear probability model Probit									
	(1)	(2)	(3)		(4)		(5)		(6)	
	Student employer entrepreneurs (1-50 workers)	Student employer entrepreneurs (1 or more workers)	Student employer entrepreneurs (1-50 workers)		Student employer entrepreneurs (1 or more workers)		Student employer entrepreneurs (1-50 workers)		Student employer entrepreneurs (1 or more workers)	
	Estimation	Estimation	Estimation	dy/dx	Estimation	dy/dx	Estimation	dy/dx	Estimation	dy/dx
Role model	0.026	0.034* (0.017)	0.197* (0.119)	0.024* (0.014)	0.242** (0.118)	0.032** (0.015)	0.217** (0.119)	0.027** (0.014)	0.259** (0.117)	0.034** (0.014)
Entrepreneur's social image	0.016 (0.019)	0.018 (0.020)	0.116 (0.130)	0.014 (0.015)	0.114 (0.128)	0.015 (0.016)	0.125 (0.131)	0.016 (0.015)	0.122 (0.129)	0.016 (0.016)
Fear of failure	-0.028*	-0.033*	-0.149	-0.019	-0.182*	-0.026*	-0.140	-0.018	-0.175*	-0.025*
	(0.016)	(0.017)	(0.106)	(0.014)	(0.105)	(0.015)	(0.106)	(0.014)	(0.105)	(0.015)
University's lack of incentives to create a new	-0.020**	-0.017*	-0.131**	-0.017**	-0.108**	-0.015**	-0.132**	-0.017**	-0.109**	-0.015**
firm	(0.009)	(0.009)	(0.052)	(0.007)	(0.052)	(0.007)	(0.052)	(0.007)	(0.052)	(0.007)
Entrepreneurial knowledge,	0.042**	0.039**	0.300***	0.039***	0.259**	0.037**	0.266**	0.035**	0.228**	0.032**
training and skills	(0.017)	(0.017)	(0.112)	(0.015)	(0.108)	(0.015)	(0.114)	(0.015)	(0.110)	(0.016)
Entrepreneurship education	0.053***	0.051***	0.351***	0.044***	0.311***	0.042***	0.362***	0.045***	0.319***	0.043***
	(0.015)	(0.016)	(0.120)	(0.014)	(0.115)	(0.014)	(0.122)	(0.014)	(0.116)	(0.015)
Income less than 1000€	0.071**	0.081***	0.446**	0.075**	0.477***	0.086**	0.459**	0.077**	0.487***	0.089**
	(0.030)	(0.031)	(0.180)	(0.037)	(0.173)	(0.038)	(0.181)	(0.037)	(0.174)	(0.039)
Income between 2000 and	0.036**	0.034*	0.245*	0.033*	0.215*	0.031+	0.239*	0.032*	0.210	0.030
4000€	(0.018)	(0.018)	(0.133)	(0.018)	(0.130)	(0.019)	(0.133)	(0.018)	(0.130)	(0.019)
Income between 4000 and 7000€	0.039	0.040	0.233	0.034	0.221	0.035	0.238	0.035	0.225	0.034
	(0.026)	(0.027)	(0.167)	(0.027)	(0.161)	(0.028)	(0.168)	(0.028)	(0.163)	(0.028)
Gender	0.040**	0.039**	0.238**	0.032**	0.218**	0.031**	0.234**	0.031**	0.215**	0.031**
	(0.017)	(0.017)	(0.109)	(0.015)	(0.106)	(0.015)	(0.109)	(0.015)	(0.106)	(0.015)
Age	0.000*	0.000*	0.011**	0.001**	0.011**	0.002**	0.009*	0.001**	0.009*	0.001**
	(0.000)	(0.000)	(0.005)	(0.001)	(0.005)	(0.001)	(0.005)	(0.001)	(0.005)	(0.001)
Constant	0.032	0.025	-2.071***		-2.050***		-2.116***		-2.085***	
	(0.050)	(0.051)	(0.349)		(0.344)		(0.344)		(0.339)	

Table 4. Estimating the probability of becoming a student employer entrepreneur

University fixed-effects	Yes	Yes	No	No	Yes	Yes
Observations	1202	1202	1202	1202	1202	1202
(Pseudo) R <sup>2</sup>	0.055[0.000]	0.051[0.000]	0.095[0.000]	0.086[0.000]	0.099[0.000]	0.088[0.000]
Probability			0.068	0.074	0.068	0.074
Log likelihood			-334.918	-350.836	-333.532	-349.758
Wald X <sup>2</sup>			56.04	51.59	58.82	53.17
AIC	389.431	450.353	693.836	725.672	693.063	725.516
BIC	455.623	516.545	754.936	786.773	759.256	791.708

p = 0.1, \* p < 0.1, \*\* p < 0.05, \*\*\* p < 0.01.Note: Heteroskedasticity corrected standard errors are shown in parentheses. *F*-statistic in brackets. Estimates for university fixed-effects dummies are not presented but can be supplied upon request.

With regards to the first hypothesis, we proposed that role models have positive effects on the probability of university students becoming employer entrepreneurs. We found support for Hypothesis 1 (due to the marginal effect of this variable), which is positive and significant in all models, especially in Model 5 (p < 0.05) and Model 6 (p < 0.05). This result is in accordance with evidence that knowing or having an entrepreneur in the family and socializing with him/her increases the probability of becoming an employer entrepreneur (Davidsson and Honig, 2003; Urbano and Alvarez, 2014). In this respect, our findings suggest that this informal institution is highly relevant, considering that our analysis is particularly focused on university students. Although most of the literature deals with the regular entrepreneurs (Kolvereid, 1996; Krueger, 1993; Maes et al., 2014; Matthews and Moser, 1995; Scherer et al., 1989), we provide new evidence in line with Guerrero and Urbano (2012), Guerrero et al. (2014) and Kirby et al. (2011), among others, who suggest that role model is a particular characteristic of the university community aimed toward the entrepreneurial activity, which is statistically significant and highly correlated with the intention of setting up a new firm. In this sense, those university students with entrepreneurial as well as job creation intention, are marginally influenced by the role models (0.027 in Model 5; and 0.034 in Model 6).

Regarding the second hypothesis, which posits that a favourable entrepreneur's social image has a positive effect on the probability of university students becoming employer entrepreneurs, the coefficient of this variable in all models is not statistically significant, and therefore, we cannot conclude anything. Although the sign is expected, H2 is not supported by the results, which is somehow in accordance with the literature. In this regard, there has been a social acceptance of entrepreneurs (Weenekers and Thurik, 1999; Acs et al., 2013). Perhaps, although this generalization exists, it could not be relevant to explain the entrepreneurial activity and specially the university student intention toward entrepreneurship. Another reason in explaining the low significance level is based on the foundations of Laužikas and Dailydaite (2015). According to these authors, entrepreneur's social image is associated with the social capital created by the entrepreneurs. In this regard, it seems that the social acceptance of the entrepreneurial success is not directly linked to the entrepreneurial decisions, but a mechanism could be required to explain its influence on entrepreneurship. In fact, Warnecke's (2014) work goes along this line, since it is suggested that the trust gained by the entrepreneurs is presumably beneficial to the access of the financial system. Here, even though our evidence is neither suggestive nor conclusive, we agree that public policies should not deviate from encouraging the trust's gains required to increase the entrepreneurial activity across individuals.

Concerning H3, which proposes that fear of failure has a negative effect on the probability of university students becoming employer entrepreneurs, we found evidence in not rejecting this hypothesis. Here, the literature suggests that fear of failure is a barrier deterring the entrepreneurial intentions of women and men (Koellinger et al., 2013; Noguera et al., 2013). In line with Cacciotti and Hayton's (2015) work, our theoretical framework suggests that this factor could be considered as context-specific phenomenon. In this regard, although the Model 5 is not

statistically significant, it has the expected sign; whereas Model 6 is significant (p < 0.1), supporting the idea that some risks in decision-making processes could generate fear in those entrepreneurs (Langowitz and Minniti, 2007; Minniti and Nardone, 2007). Additionally, our two models are shedding light on the influence of this latent variable in university students who have the intention to set up a firm and create jobs. In this context, the marginal effect found in Model 6 (and specially its significance) suggests that the probability of becoming an employer entrepreneur decreases by 0.025% when university students with the intention to create a medium or a large firm manifest to have fear of failure. This is especially useful to understand the entrepreneurial dynamics within the university context, since higher ambitions are accompanied by higher risk aversion, and therefore, less probability to become an entrepreneur.

Regarding the hypothesis about the universities' lack of incentives and their influence on the probability of university students becoming employer entrepreneurs, Model 5 and Model 6 allow not rejecting this hypothesis (p < 0.05 for both models). Our results suggest that if a university does not generate incentives through rewards, in general, students will be discouraged to engage in entrepreneurial activities that create jobs. According to Liñán et al. (2011), Guerrero and Urbano (2011) and Guerrero et al. (2014), among others, the university's rewards system is an important mechanism in providing entrepreneurial opportunities for the community. In this respect, universities should consider that the lack of this element is harmful for the entrepreneurial strategy and knowledge transfer. Precisely, our results suggest that if the lack of incentives to create new businesses is perceived by the university students, the probability of them becoming employer entrepreneurs tend to decrease whether they have the intention to create a small or medium firm (-0.017%) or even set up a large firm (-0.015%). Following the work of Rendisch et al. (2015), our results also underline the importance of creating an incentive system in order to allow the flow of knowledge from the university to the firms, whose result could increase the attractiveness of new employers in the economic sector. In this regard, by improving the rewards system, universities are not only playing a fundamental role to better the markets but also to encourage the creation of new ones (Kirby, 2006; Urbano and Guerrero, 2013).

In the case of H5, which is proposing that the entrepreneurial knowledge, training and skills have a positive effect on the probability of university students becoming employer entrepreneurs, we found support through our estimation results. Our findings suggest that if the university formally includes not only entrepreneurship subjects but also topics in the curriculum of the related areas, the knowledge, training and skills will increase in the students, and therefore, their probability will be positively and statistically modified (p < 0.05 for both models). According to Kolstad and Wiig (2015), if the university student knowledge about entrepreneurship is increased, not only an encouragement but also an achievement in terms of success could be obtained. Our findings provide evidence in this regard, since if the university students declare to have knowledge, training and skills in entrepreneurship, the probability of becoming employer entrepreneurs increases (0.035% in Model 5; and

0.032% in Model 6), which means that they are also playing a role in the economy by transferring their knowledge to workers. In this respect, universities are generating a multiplier effect through those students who are becoming entrepreneurs and are hiring employees. Additionally, our results could be suggestive in the sense that the entrepreneurial alertness is increased, along with the perception of opportunities (Estrin and Mickiewicz, 2012; Guerrero et al., 2014). Here student entrepreneurial experience or attitudes toward the entrepreneurial activity (Kolstad and Wiig, 2015).

In terms of H6, it has been posited that entrepreneurship education has a positive effect on the probability of university students becoming employer entrepreneurs. Given the results, we cannot reject the hypothesis, since this formal institution is found positive and statistically significant in explaining the employer student entrepreneur (p < 0.001 for both models). In this case, the importance of universities creating development through their student entrepreneurs is seen in the appropriate selection (and high quality) programmes of entrepreneurship (Fayolle, 2007a,b). In this respect, this is not enough with the acceptance of entrepreneurship in the curriculum, but in defining the necessary topics highly useful to conduct an entrepreneurial decision. Here entrepreneurship subjects and related topics in other areas (i.e., marketing, strategy, project management, microeconomics, etc.) should be designed from the academic government structure with a practitioner scope. The practical characteristic of those subjects could enhance the entrepreneurial attitudes in undergraduate students from the beginning (Guerrero and Urbano, 2012; Liñán et al., 2011; Louis et al., 1989). In this respect, if university students manifest to have knowledge about business plan elaboration, the probability of becoming employer entrepreneurs increases (0.045% in Model 5; and 0.043% in Model 6). Hence, our results agree with the idea that the university strategy about entrepreneurial courses should be in line with the entrepreneurship and innovation system created to accomplish sustainable regional and national economic development (Feldman, 2014; Guerrero et al., 2015).

Our control variables are also in accordance with the literature. For instance, Carree et al. (2002, 2007) suggest that the income level, at a certain point, is positively associated with the entrepreneurial activity. Our results show that the first income level (less than 1,000€) statistically increases the probability of one becoming a student employer entrepreneur. In terms of gender, Arenius and Minniti (2005) have suggested that there are differences between women and men entrepreneurs, in favour of the latter. In terms of the age, Urbano and Alvarez (2014) have argued that higher age increases the probability of one becoming an entrepreneur. In this regard, our results show a marginal effect of age increasing the probability of university students becoming employer entrepreneurs.

## 4.2. Robustness checks

We perform several robustness checks to see if our previously reported results still hold in front of a different set of variables and different econometric techniques. In particular, as noted earlier, we conduct the same model by employing three identification strategies. All these methodologies allow the perception that the magnitudes and relationships remain stable across models with little differences in both estimations and standard errors. The same occurs when the non-linear models are assessed by including a different set of variables. Comparing these models with those having all variables, the results hold. Likewise, in terms of signs, apart from the estimations results, the correlation matrix (Table 3) also shows that the relationships between the variables are as expected.

In terms of the different methods used, Models 1 and 2 correspond with the results from running a simple linear probability model through OLS regressions, while Models 3-6 are assessed through discrete choice model (i.e., probit estimations). Even though OLS regressions are inappropriate in our setting, the estimated coefficient associated with the informal and formal institutions, as well as the control variables, is still economically and statistically significant. The same is true for the binomial regression in Models 3-6, being that the results of Models 5 and 6 are highly accurate. It is reassuring that the coefficient estimates of Models 1-4 are in the middle range of the corresponding estimated coefficients presented in Models 5-6.

Regarding the different set of variables, an important observation from Table 4 is that the two dependent variables (student employer entrepreneurs hiring between 1 and 50 workers, and those student entrepreneurs hiring at least one employee and also more than 50 employees) seem to have high-predicting power regarding the effect of set variables representing informal and formal institutions. In this regard, changing the dependent variables, the model structure holds. Additionally, by excluding the university fixed-effects (Models 3 and 4), the estimation results suffer few variations, indicating once again, a robust structure.

The findings from the checks described above show that our results are stable to various changes applied to the original specification. Therefore, we are confident that the informal and formal institutions, as well as the control variables we studied, had a robust positive effect on student employer entrepreneurs.

# 5. Policy implications

The results previously described could extend the current debate about the importance of universities creating social dynamics. For instance, our results show that those variables related to the university norms have a greater impact on student employer entrepreneurs than informal institutions. This is quite interesting, since it has been argued that informal institutions tend to have greater impacts (Urbano and Alvarez, 2014). However, given the nature of the context of the UAB and UOC, it seems that students present higher response to those variables related to the university. In this regard, Audretsch et al. (2014) highlight the importance of university strategies regarding a defined rewards system, encouraging the entrepreneurial activity. Here some financial issues (i.e., scholarships, awards, financial cooperatives for university community, etc.) could enhance the knowledge transfer from universities to firms and society. According to Guerrero et al. (2015), the role of entrepreneurial universities on

regional competitiveness and growth could depend upon if there is cooperation between universities, industry and governments to encourage the innovation and knowledge-based entrepreneurial activity. This interaction between the agents could foster the economic-development process, which is particularly affected by the continuous economic crisis and increasing unemployment nowadays.

Considering the loop between the entrepreneurial activity emerged from the university context and the regional competitiveness and development (Audretsch et al., 2012), the work of Audretsch et al. (2012) suggests a balance between those public policies aimed at encouraging the entrepreneurial activity within the universities and the strategies from the regional and national systems of entrepreneurship and innovation. In this regard, societies involved in long-term knowledge dynamics should force universities to re-design their existing managerial models in order to generate an appropriate environment of collaboration with industry and government, who also should be engaged in the same long-term development project. Here, Audretsch et al. (2014) argue that financial as well as defined market structures could be created to foster the expansion of those student entrepreneurs involved in small firm projects generating jobs. Otherwise, a crowding-out effect could result if the policies and strategies do not take into account the entrepreneurial attitudes and intentions of university communities at all levels. For instance, knowledge spillover effects could be in favour only of those medium and large firms, displacing the creation of small firms from university students (Audretsch et al., 2012). In this line, Ács et al. (2014) suggest that the regional and national system of entrepreneurship should provide support in linking government, industries and society aimed toward a common purpose.

In the above respect, if university community does not transfer knowledge, a possible negative relationship between university strategies and the firm dynamics could be created (Ponomariov, 2008). Our results suggest that this negative interaction could be overcome in the long-term through either encouraging those students with entrepreneurial characteristics to set up a new firm or shaping their attitudes toward entrepreneurship. In this respect, universities should make the most of those experienced entrepreneurs, as well as incumbent firms, to convert them to other types of role models for the university students at all levels. Here Guerrero and Urbano (2012) suggest that seminars, business meetings and labs could define places for the interaction between those potential role models and the university students. Given the high correlation found between this variable and the probability of becoming a student employer entrepreneur, a positive loop of the oldest with the newest students across time could increase the entrepreneurial motivation within the university community.

According to Feldman (2014), experience and knowledge should be also obtained from those who have faced relevant decisions and failures. In this regard, universities are ideal scenarios for joining those with entrepreneurial experience to those who want to create a new venture. Feldman's work has suggested that a development process in certain places could emerge, since individuals with enough knowledge and entrepreneurial attitudes transform the developing places into new market opportunities. In this sense, universities could be that catalyst required to generate the entrepreneurial regions, constantly reinforced by the knowledge transfer and spillover from the relationship between universities and firms (Audretsch et al, 2014).

Given that our results suggest that entrepreneurship education policies and the teaching of particular entrepreneurial skills are beneficial for those university students with the intention to create a new business with employees, the quality of those strategies, as well as programmes dealing with entrepreneurship, is crucial for the entrepreneurial development. Research, teaching and innovation should stimulate the creativity of the university students toward the applicable inventions, usefulness and worth for society. Recalling that one of the main goals of universities is the production, diffusion and knowledge preservation, undergraduate programmes should adopt a new and updated view of student's profile management. This implies addressing the entrepreneurial attitudes of students, not only in terms of new firms but also in terms of new research projects. In this regard, Guerrero and Urbano (2012) point out that universities are those actors generating a specific context to promote the entrepreneurial alertness and opportunity recognition, mainly throughout educational strategies. Here, universities are not indifferent to the context in which they are involved. Taking into consideration that, for instance, entrepreneurs presumably are facing higher uncertainty, the entrepreneurial knowledge, training and skills taught within classrooms should consider that informal context affects the reality of those entrepreneurs. Hence, universities must provide the necessary information to manage the uncertainty and the subsequent fear of failure.

## 6. Conclusions

In this paper, pooled data (for the period of 2012–2015) have been used to investigate the influence of informal and formal institutions on the university students' decision of becoming employer entrepreneurs. Using institutional economics, hypotheses regarding informal institutions (role models, entrepreneur's social image and fear of failure) and formal factors (the lack of incentives to create new businesses, entrepreneurial knowledge, training and skills, and entrepreneurship education policy) were explored. Hence, the information from surveyed students of UAB and UOC was used to estimate linear probability models as well as estimate discrete choice models (i.e., probit).

In terms of the results, despite the lower statistical significance of an entrepreneur's social image, both informal and formal institutions contribute to the probability of a university student becoming an employer entrepreneur. Particularly, in this paper we found that those formal factors highly related to the university dynamics have higher influence on the student employer entrepreneurs than informal institutions. In this respect, the three formal factors assessed were statistically significant in explaining the probability. Of these three, entrepreneurship education was found most relevant in explaining the decision of becoming an employer entrepreneur. This result is in line with Guerrero and Urbano's (2012) and Fayolle's (2007a,b) works, which suggest that the entrepreneurial activity within the universities depends on the capacity to

implement high-quality programmes that provide elements such as business plan preparation or business model identification. Thus, university students could receive an appropriate training, useful to easily perceive the entrepreneurial opportunities.

In terms of the informal institutions, our results provide evidence regarding the influence of the role models and fear of failure. Both of them are highly documented in the entrepreneurship literature (Álvarez et al., 2014). In this case, role models positively explain the probability of becoming an employer entrepreneur. Concerning fear of failure (although for those entrepreneurs with 1-50 employees there is not statistical significance), for those entrepreneurs hiring at least one and also more than 50 entrepreneurs, it is likely that higher risk is perceived, and hence, fear of failure emerges as a barrier. In this case, the informal institution deters the probability of becoming an employer entrepreneur that creates large firms.

We consider that our results could be useful to extend the current debate about the importance of universities promoting development and contributing to the society. In this regard, the recognition of the 'heroic entrepreneur' is not the only factor determining the entrepreneurial success, and here, the institutions can play an important role in encouraging new ventures into fully fledged, value-adding growth ventures (Ács et al., 2014). Thus, as Williamson (2000) recognized, a policy discussion was performed, taking into account the interactions between formal and informal institutions. Basically, formal institutions, such as university norms and their government structure, must be in accordance with the environment in which university students are involved. In this sense, in the short-term, strategies related to the re-designed courses and curriculums of entrepreneurship should take into account the students' profiles, addressing them not only toward new business creation but also toward research projects identification. Additionally, a rewards system generating constant incentives must be provided to the university community. For the long-term, scenarios where experienced entrepreneurs interact with those potential entrepreneurs should be created. A reinforced loop could be achieved, enhancing the entrepreneurial attitudes and motivation toward entrepreneurship within the university community. Here, we could expect that the university system would contribute to the regional and national development by providing skilled entrepreneurs.

Our analysis also allows the identification of some limitations. First, the sample size is limited only to two universities. Although, in general, there were not important differences between UAB and OUC, future studies could consider the inclusion of the other seven universities within the Catalan system. Here, it is worth synchronizing all the universities in order to obtain higher and similar sample size to compare the informal and formal institutions that influence the university student entrepreneurs. Likewise, it would be interesting the application of the instrument in a different university system in order to compare the results. Second, given the utility of the questionnaire, other analysis could also be taken into account. For instance, the assessment of gender gap (Noguera et al., 2013) between university students could be possible, along with the influence of the university strategy in solving this issue.

Finally, due to the nature of our treated variables, probabilistic models were performed. Nonetheless, survival approach (Millán et al., 2012) could provide other insights regarding not only the analysis in terms of becoming employer entrepreneurs but also in knowing the probability of incorporating this decision in the next period. Additionally, in order to embrace higher complexity, a simultaneous equation approach would suggest how institutions (where universities are considered), entrepreneurship and jobs creation are interacting constantly (Aparicio et al., 2015). In this respect, futures research lines could extend the analysis of the university contribution to society.

#### References

- Ács, Z. J., Autio, E., & Szerb, L. (2014). National systems of entrepreneurship: Measurement issues and policy implications. Research Policy, 43(3), 476-494.
- Acs, Z. J., Boardman, M. C., & McNeely, C. L. (2013). The social value of productive entrepreneurship. Small Business Economics, 40(3), 785-796.
- Aghion, P., & Howitt, P. (1992). A Model of Growth through Creative Destruction. Econometrica, 60(2), 323-351.
- Aidis, R., Estrin, S., & Mickiewicz, T. (2008). Institutions and entrepreneurship development in Russia: A comparative perspective. Journal of Business Venturing 23, 656–672.
- Ajzen, I. (1991). The theory of planned behaviour. Organizational Behaviour and Human Decision Processes, 50(2), 179–211.
- Ajzen, I. (2001). Nature and operation of attitudes. Annual review of psychology, 52(1), 27-58.
- Ajzen, I. (2002). Perceived behavioural control, self-efficacy, locus of control, and the theory of planned behaviour. Journal of Applied Social Psychology, 32(4), 665–83.
- Aldrich, H. (1999). Organizations Evolving. London: Sage Publications.
- Alvarez, C., Ferri, E., & Urbano, D. (2011a). Observatori d'Emprenedoria Universitària OBSEU 2011. Resum Executiu. Barcelona: Universitat Autònoma de Barcelona.
- Alvarez, C., Urbano, D., Coduras, A., & Ruiz-Navarro, J. (2011b). Environmental conditions and entrepreneurial activity: a regional comparison in Spain. Journal of Small Business and Enterprise Development, 18(1), 120-140.
- Álvarez, C., Urbano, D., & Amorós, J.E. (2014). GEM research: achievements and challenges. Small Bususiness Economics, 42(3), 445–465.

- Aparicio, S., Urbano, D., & Audretsch, D. (2015). Institutional factors, opportunity entrepreneurship and economic growth: Panel data evidence. Technological Forecasting and Social Change. In press, doi: 10.1016/j.techfore.2015.04.006.
- Arenius, P., & Minniti, M. (2005). Perceptual Variables and Nascent Entrepreneurship. Small Business Economics, 24(3), 233–247.
- Audretsch, D. (2012). Entrepreneurship research. Management Decision, 50(5), 755-764.
- Audretsch, D. B. (2013). The entrepreneurial society and the role of the University. Economia Marche-Journal of Applied Economics, 32(2), 6-16.
- Audretsch, D. B., Hülsbeck, M., & Lehmann, E. E. (2012). Regional competitiveness, university spillovers, and entrepreneurial activity. Small Business Economics, 39(3), 587-601.
- Audretsch, D. B., & Keilbach, M. (2008). Resolving the knowledge paradox: Knowledge-spillover entrepreneurship and economic growth. Research Policy, 37(10), 1697–1705.
- Audretsch, D. B., Lehmann, E. E., Paleari, S., & Vismara, S. (2014). Entrepreneurial finance and technology transfer. The Journal of Technology Transfer. In press, doi: 10.1007/s10961-014-9381-8.
- Autio, E., & Fu, K. (2015). Economic and political institutions and entry into formal and informal entrepreneurship. Asia Pacific Journal of Management, 32(1), 67-94.
- Baron, R. A. (2000). Psychological Perspectives on Entrepreneurship Cognitive and Social Factors in Entrepreneurs' Success. Current Directions in Psychological Science, 9(1), 15–18.
- Begley, T. M., & Boyd, D. P. (1987). Psychological characteristics associated with performence in entrepreneurial firms and smaller businesses. Journal of Business Venturing, 2(1), 79-93.
- Bruno, R. L., Bytchkova, M., & Estrin, S. (2013). Institutional determinants of new firm entry in Russia: A cross-regional analysis. Review of Economics and Statistics, 95(5), 1740-1749.
- Bruton, G. D., Ahlstrom, D., & Li, H. L. (2010). Institutional theory and entrepreneurship: where are we now and where do we need to move in the future?. Entrepreneurship Theory and Practice, 34(3), 421-440.

- Bruton, G. D., Ahlstrom, D., & Puky, T. (2009). Institutional differences and the development of entrepreneurial ventures: A comparison of the venture capital industries in Latin America and Asia. Journal of International Business Studies, 40(5), 762-778.
- Cacciotti, G., & Hayton, J. C. (2015). Fear and Entrepreneurship: A Review and Research Agenda. International Journal of Management Reviews, 17(2), 165-190.
- Carree, M., van Stel, A., Thurik, R., & Wennekers, S. (2002). Economic development and business ownership: an analysis using data of 23 OECD countries in the period 1976–1996. Small Business Economics, 19(3), 271-290.
- Carree, M., van Stel, A., Thurik, R., & Wennekers, S. (2007). The relationship between economic development and business ownership revisited. Entrepreneurship & Regional Development, 19(3), 281-291.
- Carroll, G., and Mosakowski, E. (1987). The career dynamics of self-employment. Administrative Science Quarterly, 32(4), 570-589.
- Carter, R. B., Dark, F. H., & Singh, A. K. (1998). Underwriter reputation, initial returns, and the long-run performance of IPO stocks. Journal of Finance, 53(1), 285-311.
- Chen, C. C., Greene, P. G., & Crick, A. (1998). Does entrepreneurial self-efficacy distinguish entrepreneurs from managers?. Journal of Business Venturing, 13(4), 295-316.
- Chen, S. C., Hsiao, H. C., Chang, J. C., Chou, C. M., Chen, C. P., & Shen, C. H. (2015). Can the entrepreneurship course improve the entrepreneurial intentions of students?. International Entrepreneurship and Management Journal, 11(3), 557-569.
- Chou, T. K., Cheng, J. C., & Chien, C. C. (2013). How useful is venture capital prestige? Evidence from IPO survivability. Small Business Economics, 40(4), 843-863.
- Davidsson, P., & Honig B. (2003). The role of social and human capital among nascent entrepreneurs. Journal of Business Venturing, 18(3), 301-331.
- Degroof, J., & Roberts, E. (2004). Overcoming weak entrepreneurial infrastructures for academic spin-off ventures. The Journal of Technology Transfer, 29(3–4), 1573–7047.
- Djankov, S., Porta, R. L., Lopez-de-Silanes, F., & Shleifer, A. (2002). The regulation of entry. The Quarterly Journal of Economics, 117(1), 1-37.
- Espinet, M., Junyent, M., Amat, A., & Castelltort, A. (2015). Moving Schools Towards ESD in Catalonia, Spain: The Tensions of a Change. In Jucker, R. and Mathar, R.

(Eds.), Schooling for Sustainable Development in Europe (177-199). Springer International Publishing.

- Estrin, S., & Mickiewicz, T. (2012). Shadow economy and entrepreneurial entry. Review of Development Economics, 16(4), 559–578.
- Etzkowitz H., & Leydesdorff L. (1995). The triple helix–university–industry– government relations: a laboratory for knowledge-based economic development. EASST Review, 14(1), 14–19.
- Etzkowitz, H., & Leydesdorff, L. (2000). The dynamics of innovation: from National Systems and "Mode 2" to a Triple Helix of university–industry–government relations. Research policy, 29(2), 109-123.
- Etzkowitz, H., Webster, A., Gebhardt, C., & Terra, B. R. C. (2000). The future of the university and the university of the future: evolution of ivory tower to entrepreneurial paradigm. Research policy, 29(2), 313-330.
- Fayolle, A. (2007a). Entrepreneurship and new value creation: The dynamic of the entrepreneurial process. UK: Cambridge University Press.
- Fayolle, A. (2007b). Handbook of research in entrepreneurship education: A general perspective. USA: Edward Elgar Publishing.
- Fayolle, A., & Gailly, B. (2005). Using the theory of planned behaviour to assess entrepreneurship teaching programmes. Center for Research in Change, Innovation and Strategy of Louvain School of Management, Working Paper, 5/2005.
- Feldman, M. P. (2014). The character of innovative places: entrepreneurial strategy, economic development, and prosperity. Small Business Economics, 43(1), 9-20.
- Gnyawali, D. R., & Fogel, D. S. (1994). Environments for entrepreneurship development: Key dimensions and research implications. Entrepreneurship Theory and Practice, 18(4), 43-62.
- Grandi, A., & Grimaldi, R. (2005). Academics' organizational characteristics and the generation of successful business ideas. Journal of Business Venturing, 20(6), 821-845.
- Guerrero, M., Cunningham, J. A., & Urbano, D. (2015). Economic impact of entrepreneurial universities' activities: An exploratory study of the United Kingdom. Research Policy, 44(3), 748-764.
- Guerrero, M., Rialp, J., & Urbano, D. (2008). The impact of desirability and feasibility on entrepreneurial intentions: A structural equation model. International Entrepreneurship and Management Journal, 4(1), 35-50.

- Guerrero, M., Urbano, D., & Fayolle, A. (2014). Entrepreneurial activity and regional competitiveness: evidence from European entrepreneurial universities. The Journal of Technology Transfer. In press, doi: 10.1007/s10961-014-9377-4.
- Guerrero, M., & Urbano, D. (2012). The development of an entrepreneurial university. The Journal of Technology Transfer, 37(1), 43-74.
- Harper, D. (2003). Foundations of Entrepreneurship and Economic Development. Abingdon, Oxon, Routledge.
- Hoffmann, A., Junge, M., & Malchow-Møller, N. (2015). Running in the family: parental role models in entrepreneurship. Small Business Economics, 44(1), 79-104.
- Jayawarna, D., Rouse, J., & Kitching, J. (2013). Entrepreneur motivations and life course. International Small Business Journal, 31(1), 34-56.
- Jennings, P. D., Greenwood, R., Lounsbury, M. D., & Suddaby, R. (2013). Institutions, entrepreneurs, and communities: A special issue on entrepreneurship. Journal of Business Venturing, 28(1), 1-9.
- Jiménez, A., Palmero-Cámara, C., González-Santos, M. J., González-Bernal, J., & Jiménez-Eguizábal, J. A. (2015). The impact of educational levels on formal and informal entrepreneurship. BRQ Business Research Quarterly, 18(3), 204-212.
- Karaca, F., & Öner, M. A. (2015). Scenarios of nanotechnology development and usage in Turkey. Technological Forecasting and Social Change, 91, 327-340.
- Kickul, J., & Zaper, J. A. (2000). Untying the knot: Do personal and organizational determinants influence entrepreneurial intentions?. Journal of Small Business & Entrepreneurship, 15(3), 57-77.
- Kirby, D. A. (2006). Creating entrepreneurial universities in the UK: Applying entrepreneurship theory to practice. The Journal of Technology Transfer, 31(5), 599-603.
- Kirby, D. A., Guerrero, M., & Urbano, D. (2011). Making universities more entrepreneurial: Development of a model. Canadian Journal of Administrative Sciences, 28(3), 302-316.
- Kolvereid, L. (1996). Prediction of employment status choice intentions. Entrepreneurship Theory and Practice, 21(1), 47-57.
- Knörr, H., Alvarez, C., & Urbano, D. (2013). Entrepreneurs or employees: a crosscultural cognitive analysis. International Entrepreneurship and Management Journal, 9(2), 273-294.

- Koellinger, P., Minniti, M., & Schade, C. (2013). Gender Differences in Entrepreneurial Propensity. Oxford Bulletin of Economics and Statistics, 75(2), 213-234.
- Kolvereid, L. (1996). Prediction of employment status choice intentions. Entrepreneurship Theory and Practice, 21(3), 47-57.
- Kolvereid, L., & Isaksen, E. (2006). New business start-up and subsequent entry into self-employment. Journal of Business Venturing, 21(6), 866-885.
- Krueger, N. (1993). The impact of prior entrepreneurial exposure on perceptions of new venture feasibility and Desirability. Entrepreneurship Theory and Practice, 18(1), 5-21.
- Krueger, N. F. (2000). The cognitive infrastructure of opportunity emergence. Entrepreneurship Theory and Practice, 24(3), 5–23.
- Langowitz, N., and Minniti, M. (2007). The entrepreneurial propensity of women. Entrepreneurship Theory and Practice, 31(3), 341–364.
- Laužikas, M., & Dailydaitė, S. (2015). Impacts of social capital on transformation from efficiency to innovation-driven business. Journal of Business Economics and Management, 16(1), 37-51.
- Lee, I. H., Lévesque, M., & Minniti, M. (2012). Employees' Break-offs and Location Selection: The Birth of Industrial Clusters. IEEE Transactions on Engineering Management, 59(2), 278-292.
- Li, H., de Zubielqui, G. C., & O'Connor, A. (2015). Entrepreneurial networking capacity of cluster firms: a social network perspective on how shared resources enhance firm performance. Small Business Economics, in press. Doi: 10.1007/s11187-015-9659-8.
- Lindquist, M. J., Sol, J., & Van Praag, M. (2015). Why Do Entrepreneurial Parents Have Entrepreneurial Children?. Journal of Labor Economics, 33(2), 269-296.
- Link, A. N., & Scott, J. T. (2005). Opening the ivory tower's door: An analysis of the determinants of the formation of US university spin-off companies. Research Policy, 34(7), 1106-1112.
- Liñán, F., Urbano, D., & Guerrero, M. (2011). Regional variations in entrepreneurial cognitions: Start-up intentions of university students in Spain. Entrepreneurship & Regional Development, 23(3), 187–215.
- Loras, J., & Vizcaíno, J. (2013). Is technical training an obstacle to entrepreneurship?. Management Decision, 51(5), 999-1010.

- Louis, K. S., Blumenthal, D., Gluck, M. E., & Stoto, M. A. (1989). Entrepreneurs in academe: An exploration of behaviours among life scientists. Administrative Science Quarterly, 34(1), 110–131.
- Lukes, M., & Zouhar, J. (2015). How entrepreneurial are Czech students in higher education. In Krejci, I., Flegl, M. and Houska, M (Eds.) Efficiency and responsibility in education (325-331). Prague: Czech Univ Life Sci, Fac Econ & Management, Dept Syst Engn.
- Maas, J., Seferiadis, A. A., Bunders, J. F., & Zweekhorst, M. B. (2014). Bridging the disconnect: how network creation facilitates female Bangladeshi entrepreneurship. International Entrepreneurship and Management Journal, 10(3), 457-470.
- Maes, J., Leroy, H., & Sels, L. (2014). Gender differences in entrepreneurial intentions: A TPB multi-group analysis at factor and indicator level. European Management Journal, 32(5), 784-794.
- Mancilla, C., & Amorós, J. E. (2015). Entrepreneurship in regions: differentiated impacts of the socio cultural and gender types. Academia Revista Latinoamericana de Administración, 28(1), 45-76.
- Matthews, C. H., & Moser, S. B. (1995). Family background and gender: Implications for interest in small firm ownership. Entrepreneurship & Regional Development, 7(4), 365-377.
- Millán, A., Millán, J. M., Román, C., & van Stel, A. (2015). Unraveling the relationship between the business cycle and the own-account worker's decision to hire employees. International Entrepreneurship and Management Journal, 11(2), 321-342.
- Millán, J. M., Congregado, E., & Román, C. (2012). Determinants of self-employment survival in Europe. Small Business Economics, 38(2), 231-258.
- Minniti, M. (2005). Entrepreneurship and network externalities. Journal of Economic Behavior & Organization, 57(1), 1-27.
- Minniti, M., & Nardone, C. (2007). Being in Someone Else's Shoes: the Role of Gender in Nascent Entrepreneurship. Small Business Economics, 28(2-3), 223–238.
- Moog, P., Werner, A., Houweling, S., & Backes-Gellner, U. (2015). The impact of skills, working time allocation and peer effects on the entrepreneurial intentions of scientists. The Journal of Technology Transfer, 40(3), 493-511.

- Nielsen, S. L., & Lassen, A. H. (2012). Images of entrepreneurship: towards a new categorization of entrepreneurship. International Entrepreneurship and Management Journal, 8(1), 35-53.
- Noguera, M., Alvarez, C., & Urbano, D. (2013). Socio-cultural factors and female entrepreneurship. International Entrepreneurship and Management Journal, 9(2), 183-197.
- North, D. C. (1990). Institutions, institutional change and economic performance. Cambridge: Cambridge University Press.
- North, D. C. (2005). Understanding the process of economic change. Princeton: Princeton University Press.
- Parker, S. C., & Van Praag, M. (2010). Group status and entrepreneurship. Journal of Economics & Management Strategy, 19(4), 919-945.
- Ponomariov, B. L. (2008). Effects of university characteristics on scientists' interactions with the private sector: an exploratory assessment. The Journal of Technology Transfer, 33(5), 485-503.
- Rocha, H. O. (2004). Entrepreneurship and development: The role of clusters. Small Business Economics, 23(5), 363–400.
- Scherer, R. f., Adams, J. S., Carley, S. S., & Wiebe, F. A. (1989). Role model performance effects on development of entrepreneurial career preference. Entrepreneurship Theory and Practice, 13(3), 53-71.
- Schumpeter, J. A. (1911). The theory of economic development: An inquiry into profits, capital, credit, interest, and the business cycle. Transaction publishers.
- Scott, M. G., & Twomey, D. W. (1988). The long-term supply of entrepreneurs: Students' career aspirations in relation to entrepreneurship. Journal of Small Business Management, 26(4), 5-13.
- Stenholm, P., Acs, Z. J., & Wuebker, R. (2013). Exploring country-level institutional arrangements on the rate and type of entrepreneurial activity. Journal of Business Venturing, 28(1), 176-193.
- Stephan, U., Uhlaner, L. M., & Stride, C. (2015). Institutions and social entrepreneurship: The role of institutional voids, institutional support, and institutional configurations. Journal of International Business Studies, 46(3), 308-331.
- Stephen, F. H., Urbano, D., & van Hemmen, S. (2005). The impact of institutions on entrepreneurial activity. Managerial and Decision Economics, 26(7), 413-419.

- Thornton, P., Ribeiro-Soriano, D., & Urbano, D. (2011). Socio-cultural factors and entrepreneurial activity: an overview. International Small Business Journal, 29(2), 105-118.
- Thurik, A. R., Carree, M. A., Van Stel, A., & Audretsch, D. B. (2008). Does selfemployment reduce unemployment?. Journal of Business Venturing, 23(6), 673-686.
- Tkachev, A., & Kolvereid, L. (1999). Self-employment intentions among Russian students. Entrepreneurship & Regional Development, 11(3), 101–114.
- Trippl, M., Sinozic, T., & Smith, H. L. (2015). The role of universities in regional development: conceptual models and policy institutions in the UK, Sweden and Austria. European Planning Studies, 23(9), 1722-1740.
- Turró, A., Urbano, D., & Peris-Ortiz, M. (2014). Culture and innovation: The moderating effect of cultural values on corporate entrepreneurship. Technological Forecasting and Social Change, 88, 360-369.
- Urbano, D., & Alvarez, C. (2014). Institutional dimensions and entrepreneurial activity: an international study. Small Business Economics, 42(4), 703–716.
- Urbano, D., & Aparicio, S. (2015). Entrepreneurship capital types and economic growth: International evidence. Technological Forecasting and Social Change. In press, doi: 10.1016/j.techfore.2015.02.018.
- Urbano, D., Aparicio, S., Turró, A., & Noguera, M. (2015). Observatori d'Emprenedoria Universitària OBSEU 2015. Resum Executiu. Barcelona: Universitat Autònoma de Barcelona.
- Urbano, D., & Guerrero, M. (2013). Entrepreneurial Universities: Socioeconomic Impacts of Academic Entrepreneurship in a European Region. Economic Development Quarterly, 27(1), 40-55.
- Van Auken, H., Fry, F. L., & Stephens, P. (2006). The influence of role models on entrepreneurial intentions. Journal of Developmental Entrepreneurship, 11(02), 157-167.
- van Hemmen, S., Alvarez, C., Peris-Ortiz, M., & Urbano, D. (2015). Leadership Styles and Innovative Entrepreneurship: An International Study. Cybernetics and Systems, 46(3-4), 271-286.
- van Hemmen, S., Urbano, D., & Alvarez, C. (2013). Charismatic leadership and entrepreneurial activity: An empirical analysis. Innovar – Revista de Ciencias Adminstrativas, 23(50), 53-65.

- van Praag, M. (2011). Who values the status of the entrepreneur? In Audretsch, D., Falck, O., Heblich, S., and Lederer, A. (Eds.), Handbook of Research on Innovation and Entrepreneurship (24-42). Edward Elgar Publishing.
- van Praag, C. M., & Versloot, P. H. (2007). What is the value of entrepreneurship? A review of recent research. Small Business Economics, 29(4), 351–382.
- van Stel, A., Millán, J. M., & Román, C. (2014a). Investigating the impact of the technological environment on survival chances of employer entrepreneurs. Small Business Economics, 43(4), 839-855.
- van Stel, A., Wennekers, S., & Scholman, G. (2014b). Solo self-employed versus employer entrepreneurs: determinants and macro-economic effects in OECD countries. Eurasian Business Review, 4(1), 107-136.
- Veciana, J. M., Aponte, M., & Urbano, D. (2005). University students' attitudes towards entrepreneurship: A two countries comparison. International Entrepreneurship and Management Journal, 1(2), 165–182.
- Venkataraman, S. (2004). Regional transformation through technological entrepreneurship. Journal of Business Venturing, 19(1), 153-167.
- Wach, K. (2014). Europeanisation of Entrepreneurship Education in Europe–Looking Back and Looking Forward. Horyzonty Wychowania, 13(26), 11-31.
- Warnecke, T. (2014). The "Individualist Entrepreneur" vs. Socially Sustainable Development: Can Microfinance Build Community?. Journal of Economic Issues, 48(2), 377-386.
- Weber, E. U., & Milliman, R. A. (1997). Perceived risk attitudes: Relating risk perception to risky choice. Management Science, 43(2), 123–144.
- Wennekers, S., & Thurik, R. (1999). Linking entrepreneurship and economic growth. Small Business Economics, 13(1), 27-56.
- Wennberg, K., Pathak, S., & Autio, E. (2013). How culture moulds the effects of selfefficacy and fear of failure on entrepreneurship. Entrepreneurship & Regional Development, 25(9-10), 756-780.
- Williamson, O.E. (2000). The new institutional economics: taking stock, looking ahead. Journal of Economic Literature, 38(3), 595–613.
- Wright, M., Clarysse, B., Mustar, P., & Lockett, A. (2007). Academic entrepreneurship in Europe. Massachusetts, US: Edward Elgar Publishing.

- Zapkau, F. B., Schwens, C., Steinmetz, H., & Kabst, R. (2015). Disentangling the effect of prior entrepreneurial exposure on entrepreneurial intention. Journal of Business Research, 68(3), 639-653.
- Zhang, Y. (2015). The contingent value of social resources: Entrepreneurs' use of debtfinancing sources in Western China. Journal of Business Venturing, 30(3), 390-406.